

## **Author's response to reviews**

**Title:** Environment And Genetics in Lung cancer Etiology (EAGLE) study: an integrative population-based case-control study of lung cancer

### **Authors:**

Maria Teresa Landi ([landim@mail.nih.gov](mailto:landim@mail.nih.gov))  
Dario Consonni ([dario.consonni@unimi.it](mailto:dario.consonni@unimi.it))  
Melissa Rotunno ([rotunnom@mail.nih.gov](mailto:rotunnom@mail.nih.gov))  
Andrew W. Bergen ([bergena@mail.nih.gov](mailto:bergena@mail.nih.gov))  
Alisa Goldstein ([goldstea@mail.nih.gov](mailto:goldstea@mail.nih.gov))  
Jay H. Lubin ([lubinj@mail.nih.gov](mailto:lubinj@mail.nih.gov))  
Lynn Goldin ([goldinl@mail.nih.gov](mailto:goldinl@mail.nih.gov))  
Michael Alavanja ([alavanjam@mail.nih.gov](mailto:alavanjam@mail.nih.gov))  
Glen Morgan ([gmorgan@mail.nih.gov](mailto:gmorgan@mail.nih.gov))  
Amy F. Subar ([subara@mail.nih.gov](mailto:subara@mail.nih.gov))  
Ilona Linnoila ([linnoila@mail.nih.gov](mailto:linnoila@mail.nih.gov))  
Fabrizio Previdi ([fprevidi@craon.it](mailto:fprevidi@craon.it))  
Massimo Corno ([mcorno@impresaedilemariani.it](mailto:mcorno@impresaedilemariani.it))  
Maurizia Rubagotti ([maurizia.rubagotti@unimi.it](mailto:maurizia.rubagotti@unimi.it))  
Barbara Marinelli ([barbara.marinelli@unimi.it](mailto:barbara.marinelli@unimi.it))  
Benedetta Albettii ([benedetta.albetti@unimi.it](mailto:benedetta.albetti@unimi.it))  
Antonio Colombi ([antonio.colombi@unimi.it](mailto:antonio.colombi@unimi.it))  
Margaret Tucker ([tucker@mail.nih.gov](mailto:tucker@mail.nih.gov))  
Wacholder Sholom ([wacholders@mail.nih.gov](mailto:wacholders@mail.nih.gov))  
Angela C. Pesatori ([a.pesatori@unimi.it](mailto:a.pesatori@unimi.it))  
Neil E. Caporaso ([caporasn@mail.nih.gov](mailto:caporasn@mail.nih.gov))  
Pier Alberto Bertazzi ([pieralberto.bertazzi@unimi.it](mailto:pieralberto.bertazzi@unimi.it))

**Version:** 5 **Date:** 16 May 2008

### **Author's response to reviews:**

Dear Editor,

As requested, we are submitting again our manuscript with the authors written exactly as they are in the submission system.

Best regards,

Maria Teresa Landi, MD, PhD